Contents lists available at ScienceDirect

Information Fusion

journal homepage: www.elsevier.com/locate/inffus

A systematic literature review of low-cost 3D mapping solutions

Jesús Balado^{a,*}, Raissa Garozzo^b, Lukas Winiwarter^{c,d}, Sofia Tilon^{e,f}

^a GeoTECH, CINTECX, Universidade de Vigo, 36310 Vigo, Spain

^b Department of Civil Engineering and Architecture, University of Catania, 95123 Catania, Italy

^c Unit of Geometry and Surveying, Faculty of Engineering Sciences, University of Innsbruck, 6020 Innsbruck, Austria

^d Research Unit Photogrammetry, Department of Geodesy and Geoinformation, TU Wien, 1040 Vienna, Austria

e Royal Netherlands Aerospace Centre (NLR), Anthony Fokkerweg 2, 1059 CM, Amsterdam, The Netherlands

^f Faculty of Geo-Information Science and Earth Observation (ITC), University of Twente, 7514 AE Enschede, The Netherlands

ARTICLE INFO

Keywords: Cost-effective Solid-state lidar Photogrammetry 3D sensors Cultural heritage, Forestry

ABSTRACT

In "low-cost" solutions, ensuring economic accessibility and democratizing the availability of emerging technologies stand as pivotal considerations. This study undertakes a systematic literature review of low-cost 3D mapping solutions. Leveraging SCOPUS as the primary database, a comprehensive bibliometric analysis encompassing 1380 publications was conducted, subsequently narrowing the focus to 87 recent publications for detailed review. This research endeavors to delineate the defining characteristics of low-cost systems, elucidate their principal applications and preferred platforms, assess accessibility level, gauge the extent of innovation in both hardware and software development, explore the contributions of Deep Learning and data fusion, evaluate the consideration of data quality, and examine the contemporary relevance of photogrammetry within low-cost context. The findings demonstrate that many authors subjectively use the term low-cost to highlight qualities of a technology, methodology or sensor, but challenges arise from data quality comparisons with high-cost systems.

1. Introduction

1.1. Context

In an increasingly globalized and competitive world, the implementation of low-cost 3D mapping solutions has acquired unprecedented relevance in the industry [1]. In response to growing competitive pressure and the constant pursuit of resource optimization, companies have been forced to reassess and rethink their product and service development strategies. The prioritization of economic efficiency and consumer accessibility has taken a central place in the formulation of effective business strategies [2]. In this sense, low-cost solutions have not only allowed companies to maintain healthy profit margins in an environment of constantly increasing costs, but also facilitated penetration into previously inaccessible markets, opening new opportunities for growth and expansion [3]. This paradigm shift has led to a more inclusive and consumer-centric approach, in which economic viability is sought alongside customer satisfaction [4].

Technological advancements and the evolution of efficient production practices have primarily facilitated the transition toward low-cost solutions. Implementing cutting-edge manufacturing techniques [5], developing more affordable materials [6], and optimizing processes [7] have allowed companies to significantly reduce operational and production costs, paving the way for greater profitability and scalability. Moreover, the democratization of technology and its widespread accessibility have leveled the playing field not only for companies of all sizes [8,9] but also in different global contexts, i.e., developing countries, fostering innovation and the development of affordable, high-quality products. In this context, the strategic focus on low-cost solutions has become a crucial differentiator [10] for technology and marketing [11]. However, it is important to recognize that there is always a trade-off between low-cost, quality, and user/customer support. While low-cost solutions can make technology more accessible, they may sometimes result in compromises in product quality, or the level of customer support provided. Addressing these trade-offs is essential to give a comprehensive and objective view of the impact and potential limitations of low-cost technological solutions.

1987

INFORMATION FUSION

In this context, this study presents a comprehensive analysis of the phenomenon of low-cost solution development, with a specific focus on the field of 3D mapping. As technology has advanced and become more accessible, low-cost 3D mapping solutions have emerged as a key tool for a wide range of applications and enhancing the establishment of sensor

https://doi.org/10.1016/j.inffus.2024.102656

Received 21 April 2024; Received in revised form 9 August 2024; Accepted 27 August 2024 Available online 28 August 2024



^{*} Corresponding author. *E-mail address: jbalado@uvigo.gal* (J. Balado).

^{1566-2535/© 2024} The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY-NC license (http://creativecommons.org/licenses/by-nc/4.0/).

networks [12].

1.2. Terminology

Sensors to acquire 3D data are often grouped into active and passive sensors, where active sensors emit the energy required for the sensing themselves, and passive sensors use some other energy source (e.g., the Sun, ambient light, or thermal radiation). In the 3D domain, active sensors are typically lidar (light detection and ranging) sensors, where a direct distance measurement is carried out through time-of-flight or phase-shift methods, whereas passive sensors are typically cameras, where depth information is inferred through geometric intersection of image rays [13]. The combination of a lidar unit with a deflection mechanism and - in the case of mobile platforms - equipment to record platform position and attitude is called a laser scanner [14].

While price reduction in 3D sensors is clearly driven by the development of autonomous vehicles within the automotive industry [15,16], low-cost 3D mapping sensors are present in multiple disciplines and have given rise to new solutions and terminologies [17], for example Backpack Laser Scanning (BLS), Handheld Mobile Laser Scanning (HMLS) or Unmanned Aerial Vehicle (UAV) Laser Scanning (ULS). For a more complete overview, we refer the reader to [13]. In this manuscript, the term solution is used to indicate a method or physical device that allows 3D mapping of the environment. Consequently, a *sensor* is a single device that may be combined with other sensors into a hardware system. Sensors that are tightly coupled together (e.g., a laser scanner and an Inertial Measurement Unit (IMU)) may be considered as a single sensor. A method describes an algorithmic framework to achieve 3D mapping from some data, for example from images (photogrammetry) or laser returns (laser scanning). The literature survey (Section 2) contains contributions in terms of both low-cost solutions and methods.

1.3. Motivation

Multiple reviews have been conducted on the utilization of spatial 3D data, focusing on the analysis of specific platforms, such as Mobile Mapping Systems (MMS) [18] or indoor Mobile Mapping Systems (iMMS) [17], exploring advanced processing techniques including Deep Learning (DL) [19–21] or segmentation methods [22], or specialized applications such as Simultaneous Localization and Mapping (SLAM) for autonomous driving [23]. In contrast, comprehensive reviews concerning data acquired from low-cost sensors remain limited. In [24], the authors extensively investigate 2D Light Detection And Ranging (lidar) sensors, exploring assembly strategies and addressing the constraints associated with 3D map generation. In [25], the authors briefly provide insights into diverse types of solid-state lidar sensors, considered affordable. In [26], the authors concentrate on affordable solutions tailored for people with disabilities, highlighting the utilization of perception sensors and open-source software.

A review of low-cost sensors for 3D mapping is crucial to democratize access and promote innovation. Starting from the need to know what are considered low-cost sensors, as well as their main applications and data quality achieved, it is also important to know their feasibility to process low-cost data using Deep Learning techniques (the state of the art in 3D feature extraction and modelling), and to study the integration of lowcost sensors in more complex systems, merging information from other sensors and data sources. In addition, photogrammetry has always been a cheaper alternative to using lidar sensors, but it is important to know to what extent photogrammetry is still a viable alternative today in the face of the price reduction of 3D sensors.

1.4. Objective and research questions

The aim of this work is to address the growing need for a comprehensive and in-depth review in the field of low-cost 3D mapping solutions. As technology continues to evolve and costs remain a determining factor in the widespread adoption of these solutions, it is necessary to critically evaluate the effectiveness, feasibility, and limitations associated with the use of low-cost sensors in 3D mapping. The proposed review not only focuses on identifying emerging trends and recent innovations in this area but also aims at addressing existing gaps in the current literature and providing a holistic perspective on the applications, challenges, and potential opportunities presented by these lowcost solutions. This work will complement previous reviews and answer the following research questions:

- 1. What defines a low-cost 3D mapping sensor?
- 2. What are the primary application domains and platforms utilized for integrating low-cost 3D mapping sensors?
- 3. How accessible and user-friendly are "low-cost" 3D mapping sensors?
- 4. To what extent do current publications prioritize the development of novel 3D low-cost mapping solutions over comparative analyses?
- 5. What role does Deep Learning play in low-cost 3D mapping applications?
- 6. How is data fusion used to improve the capabilities of low-cost 3D mapping sensors?
- 7. Do low-cost 3D sensors produce poorer quality data than surveygrade 3D sensors?
- 8. What are the key differentiating factors between photogrammetric and low-cost lidar approaches for 3D mapping?
- 9. What are the current challenges in adopting and implementing lowcost 3D mapping solutions, and what future trends can be expected?

The remainder of this paper is structured as follows. Section 2 shows the methodology of the review and the bibliometric analysis. Section 3 focuses on answering the research questions previously formulated. Section 4 examines the current challenges and future trends. Section 5 concludes this paper.

2. Methodology

2.1. Search criteria

The literature review used SCOPUS as a primary data source. As one of the largest abstract and citation databases, SCOPUS encompasses a vast array of scholarly publications, conference proceedings, and books across multiple disciplines. Moreover, it provides advanced search functionalities, citation analysis tools, and the ability to track the impact of publications, thereby facilitating a meticulous and systematic review process.

The keywords used in the search were "Low-cost AND (lidar OR 3Dmapping OR 3D-sensors)" within article titles, abstracts, or keywords. These specific terms were selected due to their prominence as primary representatives within the 3D data acquisition domain. They encapsulate a wide array of mapping-centric systems, including - but not limited to - lidar sensors and depth cameras. These technologies are relevant in fields that require 3D mapping and analysis. The timeframe was limited to the period between 2018 to 2023 to focus the review on the current state of technology.

2.2. Bibliometric analysis

The proliferation of publications focusing on 3D low-cost mapping and sensors has exhibited a consistent upward trajectory over the years (Fig. 1). The initial instance of continuous search terms tracing back to 1975 highlights the mention of a low-cost dedicated processor for a lidar acquisition system [27]. Until 2010, the volume of low-cost literature remained minimal. Subsequently, there has been a significant increase. Notably, over the past decade, the term 'low-cost' has featured in approximately 3 % to 4 % of all 3D sensor-related publications. In 2023, there was a reduction in the number of publications dedicated to 3D mapping during that year and the upward trend plateaued.



Fig. 1. Evolution of the number of publications since the year 2000 according to search criteria.

Between 2018 and 2023, a total of n = 1380 documents were identified, primarily sourced from scientific journals (50.5 %) and conference proceedings (42.8 %). English emerged as the predominant language, representing 1302 publications, followed by Chinese, comprising 63 publications.

VOSviewer, a specialized software employed in qualitative research and textual data analysis, was utilized to visualize, and discern the frequency and interrelationships among keywords or terms within the textual corpus [28]. This tool generates graphical representations, enabling the identification of prevalent patterns, recurring themes, and emerging trends within the analyzed text. Specifically, the titles and abstracts of the 1380 identified publications were entered into VOSviewer for analysis. From this dataset, the 200 most frequently occurring terms and their associations were curated and visually represented (Fig. 2).

Upon analysis, VOSviewer delineated three significant clusters of relevance. The blue cluster encapsulates publications directly correlated with sensor functionalities and design, focusing prominently on technical specifications such as range, resolution, and speed. Conversely, the red cluster pertains to publications revolving around autonomous navigation, encompassing subjects like autonomous vehicles, robotics, as well as complementary concepts essential to navigation such as GNSS (Global Navigation Satellite System), SLAM, IMU and object detection. Lastly, the green cluster predominantly focuses on studies centered around measurement and monitoring methodologies, encompassing topics such as UAVs, airborne lidar, photogrammetry, and satellite application in geospatial measurement and observation.

The collection of publications showcases a diverse distribution across various subject areas, as illustrated in Fig. 3.a. Predominantly, the significant fields encompass engineering, computer science, physics and astronomy, and mathematics. Interestingly, when the search parameters are broadened beyond the specific term 'low-cost' (Fig. 3.b), a comparable trend persists in the primary focus on engineering and computer science. However, this broader scope prompts an increased relevance in the field of earth and planetary sciences within the first two subject areas, suggesting an expanding thematic emphasis in these domains.

Geographically, as illustrated in Fig. 4, the leading contributors to publications were notably affiliated with institutions from China and the



Fig. 2. VOSviewer Network analysis on terms contained in title and abstracts of the reviewed studies.



Fig. 3. Subject areas pertaining to the search categorized under 'low-cost' (a) and those not encompassing 'low-cost' (b).



Fig. 4. Distinction between countries in the context of 'low-cost' search (a) and those without 'low-cost' constraints (b).

USA, followed closely by institutions from several European nations, Canada, South Korea, Japan, and India. This distribution is explicable given that the foremost contributing institutions were predominantly China-based. Specifically, the Chinese Academy of Sciences emerged as the most prolific institution, producing 65 publications, followed by the Ministry of Education of the People's Republic of China and the University of Chinese Academy of Sciences, each contributing 38 publications, and Wuhan University with 34 publications. Following the Chinese lead, the CNRS (Centre National de la Recherche Scientifique) of France ranked next with 17 publications.

This sequence remains consistent when considering the entirety of publications on 3D sensors and mapping. However, a notable trend emerged regarding an enhanced interest among researchers affiliated with institutions from Italy towards implementing low-cost solutions. While globally, low-cost solutions constituted around 3 % of publications, Italian institutions showcased a distinct focus, with 4.5 % of their publications dedicated to low-cost solutions, indicating a specialized interest and dedication towards cost-effective methodologies within this field.

2.3. Screening, eligibility, and appraisal criteria

To streamline the comprehensive biographical review process, the initial 1380 publications underwent a significant reduction, culminating in a final selection of 87 publications. This reduction was primarily achieved by excluding publications that delved into subject areas distant from the focal point of 3D mapping, such as veterinary, medicine, pharmacology, and chemistry, among others. Additionally, publications that did not align with applications related to 3D mapping of the as-built environment, such as those centered around atmospheric measurement, were omitted from consideration.

Furthermore, publications closely associated with the fundamental principles of sensor operation and setups, as identified within the blue cluster in VOSviewer analysis, were excluded from the final selection. This decision was motivated by the existence of a recent review closely addressing this specific facet [24], rendering such publications redundant. Publications categorized as preprints or short proceedings were also omitted, ensuring a focus on publications that provided in-depth and substantive content essential for the biographical review process answering the research questions.

2.4. Research constraints overview

The primary limitations encountered during the search process can be broadly categorized into database, string, and timeframe constraints. The SCOPUS database served as the primary resource for this study, although potential future endeavors might benefit from an expansion utilizing databases such as Web of Science and Google Scholar to enhance comprehensiveness. This literature review is based solely on academic publications, excluding information from private companies and industry sources.

The utilization of a concise search string aimed to encapsulate a broad spectrum of publications aligned with the research objectives. However, this approach led to the omission of certain publications that employed techniques or sensors relevant to the study's focus but did not explicitly feature the term 'low-cost'. For instance, traditional costeffective methodologies like photogrammetry or sensors integrated into commercial devices like iPads and Azure Kinect might not always explicitly mention the 'low-cost' adjective, consequently eluding inclusion within the search results.

Moreover, the timeframe chosen for the study spanned the last 6 years, considering it as the most recent and prolific period in terms of article publications. Nonetheless, it's noteworthy that the pertinence of 'low-cost' in the context of this study extends back to around 2010. Consequently, this timeframe limitation might overlook earlier seminal works or developments related to cost-effective approaches within 3D mapping and sensor technologies.

3. Findings

3.1. Low-cost definition

In the analysis of 87 reviewed publications, only one definition of "low-cost" pertaining to 3D mapping was found. Bi et al., (2021) delineate two pivotal aspects integral to the definition of "low-cost" for 3D laser scanning:

- "Low-cost" is a relative term denoting a price significantly lower than that of comparable products, often achieved at the expense of certain specifications. This definition holds common ground beyond the domain of 3D laser scanning.
- The concept of "low-cost" may align with a consumer-grade or economically accessible product. However, the authors highlight

that existing laser scanners do not meet the criteria for being considered low-cost due to their lack of affordability for the public. They draw a comparison between the pricing of complete vehicles and lidar devices like the Velodyne VLP-16, HDL-32E, and HDL-64E. Their analysis concludes that the cost of low-cost lidar devices surpasses even the price of expensive consumer goods, such as a car, for example.

While the aforementioned definition stands as the sole delineation of "low-cost" within the reviewed literature on 3D mapping, several recurring patterns substantiate the usage of "low-cost" term across various works. The following enumeration presents these patterns in descending order of significance:

- (1) Sensors regarded as inherently low-cost without extensive justification by the authors primarily encompass devices like the Velodyne VLP-16 or VLP-32, prominent in 20 of the reviewed works [29–33], with a starting price of approximately 5000 EUR. Indeed, sensors with similar capabilities are utilized across various applications and publications without necessarily being labeled specifically as "low-cost". Additionally, other manufacturers such as Livox [34,35], SICK [36,37], Garmin [38], and SureStar [39,40] are mentioned to a lesser extent, offering devices within a comparable price range. In the realm of consumer-grade technology, the Azure Kinect [41], the Apple iPad, and Apple iPhone [42–45] are also deemed as low-cost alternatives.
- (2) Instances of low-cost techniques and technologies are highlighted in publications that do not specifically designate a sensor as lowcost (in contrast to the prior point), but rather justify a technique or technology as being economically advantageous. This pertains to:

3.2. Remote sensing, photogrammetry, or lidar: These techniques and technologies are deemed "low-cost" due to their efficiency in data acquisition compared to traditional approaches [46,47], such as total stations and manual surveys.

3.3. Simulations: Simulations are acknowledged as "low-cost" [48] as they enable data acquisition without necessitating investment in hardware or survey time.

3.4. UAVs. UAVs are considered a cost-effective alternative to traditional ALS for data collection purposes [49]. However, UAVs represent a highly specialized solution tailored for operations on a smaller scale with increased temporal resolution, rather than being inherently categorized as a low-cost solution.

3.5. Wearable devices. Although some wearable lidar devices might be more expensive than conventional devices, J. Li et al., (2022) regards wearable devices as low-cost.

- Solid state lidar [51–54] technology is deemed low-cost and encompass [25]: Microelectromechanical systems (MEMs) lidar [55], Flash lidar, Optical phased array (OPA) lidar, or Frequency-modulated continuous wave (FMCW) lidar systems.
- (2) Custom-built sensors classified as low-cost by the authors, typically tailored to address highly specific issues, such as integrating multiple sensors for urban mapping [33,56] or indoor navigation [57]. An illustrative instance is provided in [58], where the cost of a 3D lidar within the 8 to 12-meter range is stated as USD 100, while the authors' prototype is notably lower at USD 49. However, many works lack a detailed breakdown of component costs, hindering the assessment of their final pricing. Only one of the works gave a detailed breakdown of the cost of their system, at USD 12,700 [30]. A total of five publications presents the cost of the sensor they consider to be low-cost, with a range from USD 306 to USD 12,700 and a mean value of approx. USD 3500 [24, 30,34,57,58]. Direct comparisons, however, prove difficult, as

some publications give the cost of the sensor only, and others give the cumulative cost of the entire mapping system.

In broad terms, it is crucial to underscore the relative nature of "lowcost" concerning price comparisons, highlighting its extension beyond the realm of 3D mapping to align with widespread notions of affordability. While the term "low-cost" typically pertains to the initial purchase price of a device, certain authors also underscore its significance in device maintenance [59] or operational methods [60,61], particularly in utilizing free [26] or open source software [62] where the source code is either closed or modifiable and distributable, respectively. This multifaceted perspective reveals that "low-cost" encapsulates not only the acquisition cost but also encompasses the broader spectrum of expenditure and operational efficiencies within the device's lifecycle and workflow implementation.

3.2. Low-cost 3D mapping: Platforms and applications

We reviewed the publications to examine the use of low-cost 3D sensors in relation to the platforms they are integrated with and the applications they are used for. As such, we aim to understand how low-cost 3D mapping systems have evolved and have been applied in real-world settings. Only publications addressing low-cost sensors combined with a specific application or platform are considered. Those solely addressing the development of low-cost sensors or evaluating their quality are discarded. The remaining 78 publications are classified into five platforms and four application categories. These distinctive categories provide insights into the different usage of low-cost sensors across the different studies. The platform categories are manned aerial, terrestrial mobile, terrestrial static, UAV, and wearable; and the application categories are autonomous driving, cultural heritage, forestry and monitoring and inspection. Table 1 shows how often a combination of platforms and applications is addressed in this review.

3.2.1. Manned aerial

Aerial platforms, typically airplanes, are combined with lidar sensors to form an Airborne Laser Scanning (ALS) system [63]. ALS systems typically fly at higher altitudes allowing them to cover larger spatial scales and collect high-point density data. Therefore, ALS is used for forestry [47] or urban mapping and monitoring applications, such as building footprint extraction [63] or 3D building modelling [64]. Manned aerial data acquisitions are typically more costly than using other platforms. This could explain why only five publications were found using our search strategy. Low-cost in these studies refers to making the information-extraction algorithms more efficient on lower point densities, reducing the needed amount of overpasses or sensor quality, [63,64] or making manned flights obsolete by using an ALS simulator [48].

Table 1

O liter - + lie f	-1-+6	J		1	
Combination of	niattorm an	annlication	11sage in fl	ne selected	nublications
Gombinution of	piuluoini un	a application	uouge m u	ic bereeted	publications.

Platform	Application						
	Autonomous driving	Cultural Heritage	Forestry	Monitoring and Inspection	Total		
Manned aerial	0	0	1	4	5		
Terrestrial Mobile	7	1	0	23	31		
Terrestrial Static	1	0	3	7	11		
UAV	0	0	10	8	18		
Wearable	0	2	9	2	13		
Total	8	3	23	44	78		

3.2.2. Terrestrial mobile

Terrestrial mobile platforms are the most common across the 78 publications. This category can be further subdivided into various subplatforms, each with distinct uses for different applications. Several publications use lidar sensors integrated into rail trolleys or wagons for railway inspections [65,66]. Others combine them with robotic ground systems for indoor mapping [29] or with vehicles for autonomous driving applications [67]. However, most integrate them on vehicles for infrastructure inspection or asset management applications, such as traffic sign inventory measurements or road marking and pavement extraction and degradation detection [32,68-71]. Qiu et al., (2023) are the only authors to use depth sensors (Azure Kinect) for pavement monitoring. A vehicle equipped with lidar sensors for inspection purposes is commonly referred to as a Mobile Laser Scanning (MLS) or Mobile Mapping System (MMS). However, these terms are used interchangeably to denote a variety of non-vehicle platforms, such as those mentioned here or wearables (Section 4.5).

3.2.3. Terrestrial static

All publications in our review equip static terrestrial platforms with a lidar sensor to form a Terrestrial Laser Scanning (TLS) system. None equip them with photographic systems. The TLS systems are used for pavement inspection, high-definition map creation, or traffic monitoring [72–74]. Other authors use TLS systems for deriving forest structure parameters [75]. A special type of static terrestrial application includes those that fix 3D sensors to static structures. An example can be found in [73], where a lidar sensor is fixed alongside a road to measure the distance between vehicles moving in traffic.

3.2.4. Unmanned aerial vehicles (UAVs)

Typical use cases for low-cost 3D mapping with UAVs are forestry, cultural heritage or monitoring and inspection applications. UAVs are an ideal platform for carrying out monitoring of infrastructures that cover large linear infrastructures, such as roads [76], or for inspecting (industrial) assets or forests that are hard to reach or that need to be inspected from various perspectives that are outside the field of view of humans [77]. Laser sensor on UAVs have been investigated for 3D forest mapping by [30] and for 3D UAV bridge inspection by [78]. Surprisingly, our search strategy (see section 2.1) returns zero publications that use UAVs for low-cost cultural heritage applications. However, when adding the terms "UAV" and "cultural heritage" and removing "low cost" to the search term, more examples of low-cost photogrammetric 3D bridge mapping and cultural heritage applications are found. This suggests that the domain of cultural heritage is operating from a low-cost baseline and therefore does not emphasize the word low-cost in the manuscript or its tags. Examples of photogrammetric 3D bridge mapping and cultural heritage applications can be found in [79-83].

3.2.5. Wearables

A relatively novel platform is wearables. A wearable system allows low-cost 3D sensors to be carried in hand, such as Apple devices (iPhone or iPad) or attached to helmets or backpacks. Examples include helmets in industrial monitoring applications [50], assistance technologies for disabled persons [26] or custom-made backpacks for indoor 3D mapping [84,85]. However, most publications addressing wearables use commercial devices for cultural heritage or forestry applications [35,85–88]. The generally constrained budget for cultural heritage projects makes these affordable and accessible devices especially appealing although the level of detail is not always satisfactory compared to high resolution TLS data [86].

3.2.6. Platforms and applications over time

Table 2 shows how often applications or platforms were addressed between 2018-2023. Most notable is the sharp increase in monitoring and inspection applications for infrastructure and urban environments. Several factors could increase the need for automated and large-scale

Table 2

Occurrence of publications grouped by application and platform in selected publications over the review period.

Year						
2018	2019	2020	2021	2022	2023	Total
0	0	2	3	2	1	8
0	0	0	2	0	1	3
2	3	3	6	6	3	23
2	1	5	4	12	20	44
0	1	2	1	0	1	5
1	1	3	4	7	15	31
1	0	1	1	4	4	11
2	2	4	4	5	1	18
0	0	0	5	4	4	13
	Year 2018 0 0 2 2 0 1 1 2 0	Year 2018 2019 0 0 2 3 2 1 0 1 1 0 2 2 0 0	Year 2018 2019 2020 0 0 2 0 0 0 2 3 3 2 1 5 0 1 2 1 1 3 1 0 1 2 2 4 0 0 0	Year 2018 2019 2020 2021 0 0 2 3 0 2 3 3 6 2 2 3 3 6 2 2 1 5 4 0 1 2 1 1 1 3 4 1 0 1 1 2 2 4 4 0 0 0 5	Year 2018 2019 2020 2021 2022 0 0 2 3 2 0 0 2 0 2 0 2 3 3 6 6 2 1 5 4 12 0 1 2 1 0 1 1 4 2 2 4 4 5 0 0 0 5 4	Year 2018 2019 2020 2021 2022 2023 0 0 2 3 2 1 0 0 2 0 1 2 3 3 6 6 3 2 1 5 4 12 20 0 1 2 1 0 1 1 1 3 4 7 15 1 0 1 1 4 4 2 2 4 4 5 1 0 0 0 5 4 4

monitoring. Aging and degrading infrastructures and increased traffic volumes and loads lead to accelerated degradation of infrastructures. Climate change-induced prolonged weather extremes have a similar accelerated degradation effect [89]. Another reason for the increased interest in monitoring and inspection applications could be the increased computational capacity that is available or the relative ease and low-cost at which data can be collected. This presents the opportunity for urban planners to start scanning their assets from an asset's design to construction and, eventually, the service life phase. Digital twinning requires extensive 3D information collection and storage at various stages of an asset's life cycle and is increasingly becoming standard practice. Integrating Building Information Modeling (BIM) into this process further enhances the precision and utility of digital twins by providing a detailed digital representation of the physical and functional characteristics of a building or infrastructure, facilitating better decision-making throughout the entire lifecycle [90].

Forestry is a domain that has seen a steady usage of low-cost 3D sensors due to the scale at which forestry studies are usually conducted. Especially interesting is the recent rise of wearables. Low-cost aerial mapping using UAVs is traditionally appealing because the spatial scale at which forestry studies need to be conducted would be too expensive to carry out at large scales without aerial remote sensing platforms. Now the usage of wearables seems driven by the need of practitioners to obtain detailed 3D metrics at tree stand level, which can inform them about their forest's health. In addition, there is a rising consensus that highly detailed laser systems that provide millimeter-level resolution are overdone and that metrics such as Diameter at Breast Height (DBH) can also be derived from lower-resolution data, making wearables more appealing [87].

We see a steady but small number of autonomous driving applications that use low-cost sensors. Automotive applications require relatively high-detail data, something that low-cost sensors cannot provide [86]. Similarly, we find a relatively low number of cultural heritage applications in the review period. This could be explained by the prevalence of survey-grade sensors, which were already widely used in cultural heritage studies.

Regarding the types of platforms, it is noticeable that traditional platforms, such as static or airplane platforms, are steadily used. Surprisingly, the number of publications using UAVs in combination with low-cost 3D sensors is stable for the reviewed period. UAVs have the disadvantage of being sensitive to vibrations and weather conditions, which influences the range of errors in their 3D measurements and makes them unappealing to be used for precise measurements. The usage of terrestrial and mobile systems increased. Mobile terrestrial solutions are relatively low-cost for applications such as road and rail inspections where repeated detailed measurements are required at large spatial scales. It is interesting that although terrestrial platforms are traditionally used for applications where high-resolution data are needed, both mobile and static terrestrial platforms are used to address similar monitoring and mapping applications in the industrial, cultural heritage and forestry domains. While some authors advocate for using high-resolution TLS data to characterize environments such as forest features [36], others aim for an efficient trade-off between data resolution and data acquisition speed using wearables.

Overall, the publications rarely mention the effectiveness or generalizability of their method over different study sites, let alone geographic regions with different visual characteristics. We conclude that the variety and range of applications covered in the publications are often at lower development levels and not mature enough to draw conclusions on their usability and scalability in real-world settings.

3.3. Low-cost accessibility and user friendliness

As shown above, our review covers a diverse range of low-cost 3D mapping systems, some commercially fabricated, and some custommade. The way in which data are recorded and processed is distinctively different for both types.

For most commercial systems, the way hardware and software interface and the way data are recorded, processed, or presented to the customer is determined by the manufactures. An example is a commercial wearable system. In this review, they often make use of RGB-D sensors, e.g. Azure Kinect, Zed2 or Apple sensors. Azure Kinect and Zed2 sensors are accompanied by proprietary developer kits to process data, which are free and open source. Data acquired by Apple devices can be processed with third-party applications such as Sitescape or AppleAR kit which are not always free or open source. Commercial TLS systems in this review are Trimble or Riegl-based systems. Although their data can be retrieved by free software, such as RiVlib¹, processing them requires using paid and closed software that is proprietary to these brands. Commercial UAV platforms in this review mostly use DJI systems. Some of them are sometimes customized by authors themselves or by third parties [30,34,40,91]. Other commercial reviewed brands are Zenmuse [91] and Applanix [92], however, these are used less than DJI drones.

Custom-made wearables are for example helmet [35], backpack [84], or railway inspection systems (Rahman, 2023). These solutions have in common that they are built as robotic systems. Data recording and hardware-software interfacing are achieved using the free and open-source Robotic Operating System (ROS)². An exception is custom built UAV solutions where data collection and system integration is achieved using on-board computers, such as PX4³ [93].

Most UAV mapping systems use the commercial software Pix4D⁴ to perform Structure from Motion (SfM) [79,82,91,94]. In addition, most lidar-based system use commercial software to process data, such as LiAcquire and LIDAR360⁵ (Hu, 2020) or MATLAB⁶ scripts. CloudCompare⁷ is the only free and open-source software that is used across all types of systems to classify or filter point cloud data [65].

Finally, most publications do not mention what training is needed to operate the system, what the user experience is while using the system, or publish their code or solutions online. Only [87] explicitly state that Apple systems provide the best user experience. Nonetheless, it is presumed that a high level of training is required due to the complexity of the workflow. Each stage, from acquisition to processing, requires a deep understanding of the system properties or of which parameters work best [66].

3.4. New low-cost research methods

The reviewed publications were categorized based on whether their primary objective presents a novel contribution in terms of methodological development (software and/or hardware) or if the publications are centered around technical comparisons between mapping solutions. The quantification of publications is depicted in Fig. 5, followed by an explanation of the findings for each category.

- Development of new algorithms: This category encompasses 34.5 % of publications and focuses on the development of novel software, typically centered around feature extraction, object detection and segmentation [88], as well as complex algorithmic developments for data fusion from multiple sensors [95] or simulations [48].
- Development of new systems: This category comprises 24.1 % of publications and focuses on descriptions of hardware setups and sensor calibrations that collectively provide a new system for 3D low-cost mapping [58]. The complexity of the setups varies depending on the application. The sensors are usually installed on a vehicle [37], robotic bases [96] or UAVs [30]. These publications are complemented by visualizations of the captured data and quality analyses.
- Development of new systems and algorithms: This category comprises 23.0 % of publications and provides complete mapping solutions to problems, from the design and installation of low-cost sensors to calibration and software development [32,70,97,98].
- Comparative quality analysis: This category encompasses 16.1 % of publications and compares different low-cost sensors, systems, or mapping solutions within their respective case studies. Some of these publications focus on analyzing the data quality [45,46,99], while others compare methodological developments using low-cost data [55]. It is also common practice for these comparisons to be conducted by juxtaposing consumer systems, with the iPad Pro being the most frequent [42,43], either with other low-cost devices or with well-established solutions in the field. Recently, comparative analysis went beyond technical specifications and also compared different apps (software) on commercial systems, since apps also influence data acquisition and quality [100].
- Survey: This category includes 2.3 % of publications and consists of reviews without conducting studies or comparisons on real use cases.

Overall, the development of new methods (software and/or hardware) accounts for 81.6 % of the reviewed works. To assess the quality of these new methods results are often compared against survey-grade equipment, with TLS being the most representative [40,44,65].

3.5. Deep Learning in low-cost 3D mapping

The representation of Artificial Intelligence (AI) applications in the low-cost 3D mapping domain is limited. Most of the low-cost literature continue to use heuristic methodologies [37,63,71,101–104]. Only 13 works employ AI in segmentation or classification tasks. In [66], Machine Learning is integrated within a heuristic framework, specifically the unsupervised clustering algorithm k-means. Prioritizing the use of heuristic methodologies makes sense since AI-based approaches, especially DL, require significant computational resources for training and implementing Deep Neural Networks (DNN), resulting in higher operating costs. In addition, most heuristic approaches can be tuned to better suit a specific application by using threshold parameters. However, the risk is that these threshold values are typically set by hand, which makes it difficult to extrapolate, generalize or transfer the developed heuristic method to other geographic areas or tasks, even if they are similar.

Currently, the trend in DL is to employ end-to-end approaches with two input options: 2D images or 3D point clouds. Image-based DL has been around for a much longer time due to a larger interest from the computer vision community and reduced computational complexity.

¹ http://www.riegl.com

² https://www.ros.org/

³ https://px4.io/

⁴ https://www.pix4d.com/

⁵ https://www.greenvalleyintl.com//software

⁶ https://www.mathworks.com/products/matlab.html

⁷ https://www.danielgm.net/cc/



Fig. 5. Distribution of "low-cost" publications according to the development of new research methods.

Hence, DL approaches working directly on the 3D data are still sparse, as e.g. the first methods were developed not more than 8 years ago [105]. Therefore, approaches dealing with image manipulation, obtained either from cameras, or from rasterizing point clouds are currently typically more efficient. In the literature survey, many applications of DL focus on object detection [67,68,106] or semantic segmentation [41, 56,69,107,108].

Applying DL directly to point clouds without rasterization is the alternative to using 2D input data. A U-Net-styled architecture built upon KPFCNN can segment objects in UAV lidar point clouds [54]. Point Transformer is another alternative, used for road inventory, although with lower accuracy rates in traffic sign classification compared to 2D-CNN [70].

Another crucial aspect is the creation of datasets for DL training and testing, as this process is laborious and costly. In most reviewed works, there is no low-cost strategy presented for data labeling, and this process is entirely manual [56,68]. Nonetheless, the computer vision domain has seen a growing interest in DL-assisted labeling approaches, using for example one-shot learning [109] or Segment Anything [110]. An exception is found in [47] where the samples were generated through Object-Based Image Analysis using thresholds in a Canopy Height Model (CHM) and the Normalized Difference Vegetation Index (NDVI).

In this context, low-cost works prioritize methodologies that require lower computational cost for data analysis. Among these methodologies, the most efficient 2D-CNN are the most relevant, such as Fast-SCNN, as they provide superior and faster results than using raw point clouds as input data in end-to-end approaches. While for post-processing tasks, computational power is less critical and is available through big providers in the cloud and at reasonable prices, online and real-time processing is more of an issue. This is especially true for low-cost systems, where expensive, high-throughput hardware like GPUs is not a feasible option or where they make use of computationally constrained edgedevices such as the NVIDIA Jetson's or the Raspberry Pi.

3.6. Data fusion in low-cost devices

Throughout the investigated publications, two main types of data fusion manifested: The fusion of imagery and lidar data (Section 3.8), and the fusion of lidar and IMU data. In the latter case, a SLAM approach is often used to find both the trajectory of the system and the resulting point cloud simultaneously [111]. IMU data are hereby used to aid in the trajectory estimation (localization) [112], especially in difficult geometries where the SLAM approach may be ambiguous.

Data fusion can - in general - be applied on multiple levels [113]: as a tight integration such as in SLAM approaches [114], where the original observations of the different sensors are combined to get an improved result [52,53,95,97], or at a later point, where information is first extracted from the data, and later fused together (loose coupling). Tight integration often makes use of the Kalman filter, which allows the combination of different measurements at different rates to obtain an estimate of, e.g., position [38,74,85]. The sensors used for data fusion

are often common cameras, and -in the 3D mapping domain- laser scanners. However, these scanners are less common and, therefore, also comparatively expensive, albeit having a much lower price-tag than survey-grade laser scanners.

Many of the loose fusion approaches presented in the publications use DL approaches for tasks like image classification before carrying out data fusion, typically with lidar [68,96,106]. Data fusion approaches combined with low-cost sensors attempt to compensate for the generally lower - data quality of these sensors. Especially in non-optimal conditions the quality of the result is improved significantly by this fusion. For example, [38] show variances in lidar and IMU data were reduced by almost a factor of two. [53] achieved similar results, with decreases in positioning Root-Mean-Square-Error (RMSE) by up to 70 % when fusing GNSS, IMU, and lidar data.

A similar concept to data fusion is Augmented Reality (AR), where data are overlaid on a real-time video stream to enhance the video with information. While AR is often reported as the main application for consumer low-cost sensors, such as the Apple iPad Pro [42,43], only one publication developed a tool where annotations and measurements can be input on the 3D sensed data and experienced through a smartphone display [88].

In summary, by combining multiple sensors in one system and fusing the data streams, developers of low-cost systems have managed to use comparatively cheap hardware like cameras [115] and IMUs and combined it with less common hardware like low-cost laser scanners. This combination allows either a transfer of methods existing for imagery and mapping it onto 3D data, or an improvement of the quality of the 3D data.

3.7. Low (cost) data quality

Data quality is an important topic, especially when discussing lowcost sensors. Three different groups were identified with regard to data quality management: (a) publications that investigate the quality of the target variables for a specific use-case (e.g., diameter at breast height (DBH) in forestry, [34]), often in comparison to survey-grade systems, (b) publications that repeat datasheet specifications and deduct some quality information, and (c) publications that do a full in-depth analysis of data quality and consider the - typically lower - quality of low-cost sensors in their method or analysis [38,40,116]. A small number of publications did not discuss data quality at all.

Data quality considerations also very much depend on the sensor type that is being investigated. When sensors are combined in more complex systems, e.g., a laser scanner with an IMU, data quality is more often discussed in terms of the final product than the individual components. For example, in a cultural heritage application, quality is investigated as mean surface roughness over the whole point cloud [60]. Furthermore, the discussion of quality also aligns with the intended use-case. For example, applications in forestry often discuss the obstruction of signals (GNSS or laser) by the canopy [30,117].

Overall, there is a consensus that more expensive survey-grade

equipment is considered to have better data quality and can be used as a reference for low-cost systems, especially ones that combine several low-cost sensors [69]. Data quality is investigated in terms of (ranging and positioning) accuracy, occlusions, and data density. Most commonly the applicability of a method or system is evaluated for a specific use case, i. e., *fitness for use* [46]. While this is a valid approach when solving a specific problem, it makes comparisons between different methods, systems, and sensors difficult. Typical results of the investigations are that a certain system's quality suffices for a specific selected use-case.

When numeric comparisons are carried out, the quantification of a RMSE is prevalent. Fewer studies try to quantify precisions and bias values separately, only if they find a physical or geometric reason to do so [42,94].

Some publications compare different methods, such as laserscanning point cloud with one obtained by photogrammetric dense image matching [49]. While these comparisons are indeed essential, it is important to account for the inherent differences in point density and occlusions caused by the sensing methods themselves. If these factors are not properly considered, the comparisons may become biased, leading to potentially misleading conclusions [118].

Publications that present the use of low-cost sensors together with DL approaches tend to quantify data quality by detection performance, i.e., the ability to detect objects of interest [29,67,106]. Accuracy, a term typically used to describe the distribution of differences from the measured values to the reference value, is then calculated by the ratio of correctly detected objects to all objects. Similarly, precision, commonly describing the repeatability of a single measurement, is the ratio of true positive detections to all positive detections. Consequently, when comparing such values across different publications, as one may give a precision of 3 mm (e.g., as a standard deviation), and another may give a precision of 95 %.

In general, there is little to no consensus on the usage of metrics for data quality, and authors tend to use metrics adapted to their field of study. Similarly, the use of terminology is not standardized and may lead to confusion when attempting direct comparisons. Publications that do attempt to characterize data quality allow experienced users to evaluate whether a sensor could be used for their use-case, albeit the transfer of such data quality considerations remains difficult. Interestingly, authors rarely mention a lack of intercomparability to other methods. While there is consensus among the publications that low-cost sensors typically exhibit lower data quality, most investigations merely show that it is sufficient regarding their application and do not quantify the difference to survey-grade equipment. If provided, datasheet values of sensor variances present parts of this reduced quality.

3.8. Photogrammetry vs low-cost lidar

Photogrammetry, particularly Multi-View Stereo (MVS) matching, and lidar represent two divergent paths in the realm of 3D surveying and documentation, each offering distinct advantages and drawbacks [119]. Photogrammetry, known for its cost-effectiveness and accessibility, leverages mainstream technology like digital cameras and smartphones. This democratizes the field of 3D surveying by enabling a broader audience to participate in digital documentation processes, without the need for specialized equipment. Additionally, the user-friendly nature of many photogrammetric software packages, i.e. Agisoft Metashape⁸, Meshroom⁹, 3DF ZEPHYR¹⁰, Reality Capture¹¹, Autodesk ReCap Pro¹², which often include intuitive, plug-and-play functionalities, further lowers the barrier to properly performing photogrammetric acquisition

and processing. These characteristics make the photogrammetric approach especially valuable in fields where budget constraints frequently limit documentation efforts.

Especially, in educational settings, photogrammetry facilitates hands-on learning by allowing students to use their personal devices for digital surveying exercises [120,121]. This approach is particularly beneficial in cultural heritage preservation, where the detailed 3D modeling of sites and artifacts is crucial yet often hindered by financial limitations. However, it is worth noting that, despite its accessibility and cost-effectiveness, photogrammetry is not universally applicable. Specific applications, in the forestry domain [122], underground archeology [123], or the presence of narrow spaces [124] or water bodies [125], are challenging for photogrammetric techniques due to the limitations in lighting conditions, and the need for discernible texture in the environment, such as the uniformity encountered with snow-covered landscapes [126].

Comparing photogrammetry to both conventional lidar and recent lidar-enabled devices like the iPad, it becomes clear that while photogrammetry can be more precise under the right circumstances due to its adaptable ground-sampling distance, lidar offers advantages in terms of intuitiveness and ease of use. For instance, lidar sensors, including those available for tablets, provide intuitive interfaces and can capture accurate 3D data even in challenging environmental conditions. On the other hand, photogrammetry offers high-resolution color textures and detailed visuals that low-cost lidar devices cannot, making it suitable for applications requiring detailed visual accuracy [127].

The integration of terrestrial photogrammetry with UAV imaging exemplifies an advanced approach to overcoming some of photogrammetry's limitations, enabling effective data capture from inaccessible or elevated locations.

In addition to photogrammetry, videogrammetry—where video footage is used to produce 3D models— remains a valuable tool in the digital documentation toolkit [128]. Videogrammetry can complement traditional photogrammetric techniques, offering dynamic data capture capabilities that are particularly useful in documenting large-scale events or moving objects [129]. Moreover, videogrammetry allows to shift work towards the post-processing phase, by acquiring videos during the survey and then selecting the frames for photogrammetry at a later stage, although this carries setbacks related to larger data storage requirements and increased processing time.

Photogrammetry remains a viable and accessible alternative, or valuable integration [130,131], to lidar for 3D surveying and documentation. Its cost-effectiveness and user-friendly interfaces support widespread use in both educational contexts and preservation projects. However, acknowledging its limitations and the necessity for integrated approaches in certain scenarios is crucial. As the field evolves, ongoing improvements and the inclusion of techniques like videogrammetry are essential for expanding photogrammetry's application range.

4. Discussion: Current trends and future challenges

4.1. Accessibility and affordability

The price and accessibility of low-cost solutions are still far from being accessible for the average consumer. The market lacks solutions specifically designed, marketed, and distributed for personal, familial, or household utilization, barring certain exceptions like Apple devices, which, while versatile, are not primarily dedicated to 3D mapping. Moreover, a substantial portion of systems remains overpriced, ranging from USD 300 [58] to USD 13,000 [30] and it is difficult to obtain complete information on prices and technical specifications without contacting distributors directly. This lack of transparency inhibits the democratization of remote sensing, where lower costs could serve as a catalyst for increased accessibility and adoption of novel techniques.

⁸ https://www.agisoft.com/

⁹ https://alicevision.org/

¹⁰ https://www.3dflow.net/

¹¹ https://www.capturingreality.com/

¹² https://www.autodesk.it/products/recap/

4.2. Mobile platforms and sensor networks

The increased usage of mobile platforms coincides with the increased number of monitoring applications, especially in the industrial domain (infrastructure monitoring and asset management). Rapidly deteriorating infrastructures, which drives the need for highly automated, precise, and scalable monitoring solutions [89]. Mobile platforms provide the flexibility and replicability that these applications require. These same needs could explain the increased interest in wearables. They are user-friendly and readily replaceable. The forestry sector is increasingly interested in wearable technology, prioritizing rapid and convenient data collection over high-resolution data. In the same vein, fast measurements using wearables, mobile platforms or UAVs can provide a tremendous advantage in applications such as disaster mapping. Surprisingly, we see no publications covering this type of application in this review.

The usage of UAVs or manned aerial platforms remained steady over time, and we expect this trend to continue in the near future. We see a steady increase in low-cost sensors embedded into mobile platforms. This could be motivated by the intersecting domains of low-cost sensors and the Internet of Things (IoT) [33,53,132]. IoT increases the intelligence of mobile platforms by providing computing power, sometimes attuned with GPU edge-devices for DL applications, or by making them connected to a communication network. In this way, mobile platforms form a comprehensive system that can process and transfer data faster and more efficiently.

4.3. Open repositories and Deep Learning software

Low-cost mapping systems use a diverse range of closed and opensource software. Almost all systems utilize a piece of closed or proprietary software in their. There is no clear indication that the publications rely on the code of other authors in the scientific community nor do authors standard publish their code under open science practices. As such there is no evidence of an existing "community of best practices".

The low-cost mapping systems reviewed here employ a variety of closed and open-source software with almost all utilizing some proprietary software. We believe this explains why the code on which they are built is not shared on public repositories. While ideas are shared in the publications themselves, it is difficult for outsiders to compare systems or learn from the system's building blocks.

Significant advancements have been made in the development of novel neural network architectures tailored for point cloud data. However, despite these strides, the integration of point-based neural network architectures into low-cost data solutions remains largely unexplored. There is a noticeable gap in the attention given to enhancing automatic labeling processes, which is extensively researched in computer vision [133]. Open access initiatives and sharing code repositories have not been seen as a common practice in most publications, although collaborative efforts could accelerate progress in low-cost DL 3D mapping solutions. Additionally, the absence of dedicated low-cost hardware solutions for real-time DL applications poses a challenge. While edge GPUs and Tensor Processing Units (TPUs) have not yet surfaced prominently in low-cost literature, their potential incorporation hints at a prospective avenue for future research [134].

4.4. Data fusion with deep learning

Data fusion can be grouped into loose and tight coupling, where - in our context of low-cost sensors - loose coupling typically fuses data derived from imagery (e.g., class labels or detection results) with lidar data. Tight coupling is observed mostly though Kalman filters in location estimation using IMU and lidar data. Often, the combination ensures appropriate data quality even if one of the sensors provides degraded results, e.g. due to environmental conditions. This is especially common with GNSS devices. In the future, the use of neural networks for data fusion may solve current issues where a tight coupling is currently not possible due to a missing physical model, and a loose coupling does not fit the application. The use of neural networks for data fusion has been shown to be successful [135], and largely depends on available training data. In geospatial and 3D mapping applications, training data can be generated in large quantities using survey-grade instruments, directly observing the targeted high-quality results.

4.5. Need for comparative analyses

Many low-cost 3D mapping papers concentrate on the development of new hardware and software, which is propelled by two primary factors. Firstly, the specific demands generated necessitate the creation of novel systems to accommodate the 3D environments. Secondly, the unique properties of data captured by low-cost devices require the continual development of specialized algorithms to effectively harness this data for mapping purposes. However, there exists a notable scarcity of works solely dedicated to comparative analyses between different solutions, whether commercial or non-commercial, or comprehensive literature reviews. Such comparative studies and reviews play a crucial role in elucidating the strengths and weaknesses of existing solutions, guiding future research directions, and facilitating informed decisionmaking for practitioners and researchers alike.

A major challenge in evaluating the data quality in general is the difficulty in comparing given metrics. Even if the metrics are the same (e.g., RMSEs to a higher-order reference), the study objects differ by the required quality and how accurately they can be sensed. For example, the requirements for a sensor for documentation of cultural heritage are very different from those for applications in forestry. This is an issue for both low-cost and high-cost systems; and especially for the comparison between the two, where the comparability between sensor datasheets is further hindered by the fact that manufacturers give values adapted to their main customer base or use case (e.g., forestry vs. structural engineering in laser scanning). A solution to this could be a common, easily recreatable benchmark. While a 3D-printable object may offer such opportunities on small scales [136], the larger scales typically of interest for low-cost 3D topographic mapping tasks, make such an endeavor difficult.

4.6. Evolution of photogrammetry

In today's rapidly evolving digital landscape, photogrammetry remains a viable and highly accessible option for generating detailed 3D models, thanks to its straightforward application and cost efficiency. Recognizing this continuous evolution, photogrammetry is increasingly complemented by advancements such as learning-based MVS [137-139] and Neural Radiance Fields (NeRF) [140]. While learning-based MVS aims to create depth maps through CNN approaches [141], NeRFs create 3D scenes from a limited set of images with an unparalleled level of detail and realism [142]. Another current trend is Gaussian splatting [143] which offers an innovative approach to 3D scene rendering and reconstruction. A smooth, continuous surface is produced by treating each point in the cloud as a multidimensional Gaussian entity and merging these points. This potentially address traditional photogrammetry challenges, such as effectively handling transparent or reflective surfaces. These advancements indicate that the domain of photogrammetry and its related fields is not just keeping pace but is at the forefront of blending traditional methods with the latest technologies to enhance and expand the capabilities of 3D modeling and virtual reconstructions.

5. Conclusions

This study has comprehensively examined low-cost 3D mapping solutions to shed light on their defining characteristics, applications, accessibility, technological advancements, and potential future directions. Through a systematic literature review and bibliometric analysis of a substantial number of publications, we can conclude that:

- Authors use the term "low-cost" to highlight that a sensor, system, or solution has a lower cost than an equivalent one, typically referred to as "survey-grade".
- Various low-cost mapping systems and applications have been identified. Some publications on low-cost mapping are driven by the need for automated and improved monitoring, while others are simply exploring the use of increasingly available low-cost commercial platforms for their specific domains.
- All publications use, to some extent, a closed or proprietary piece of software in their system, making it increasingly difficult to replicate or compare solutions, especially in a low-cost manner.
- Although 81.6 % of the publications on low-cost 3D mapping present new hardware and software developments, comparisons of commercial systems are also relevant since they are much more accessible and attract more interest from potential buyers.
- The use of DL in low-cost 3D mapping focuses on end-to-end approaches with 2D images or rasterized point clouds and prioritizing efficient CNN for both 3D reconstruction and semantic labelling.
- Data fusion is employed on different levels in low-cost systems to leverage affordable cameras or IMU systems for enhancing results and augmenting the capabilities of more expensive sensors such as lidar units, thereby improving data quality.
- High-cost sensing systems discuss data quality, while low-cost systems often overlook it, leading to performance evaluations based on final reports or DL metrics, and making comparisons challenging due to the assumption of low-cost data adequacy.
- The integration of photogrammetry into low-cost 3D mapping solutions highlights its central role in democratizing the field, enabling widespread engagement in detailed digital documentation and modeling using common technology.

CRediT authorship contribution statement

Jesús Balado: Writing – review & editing, Writing – original draft, Methodology, Investigation, Conceptualization. Raissa Garozzo: Writing – original draft, Investigation. Lukas Winiwarter: Writing – review & editing, Writing – original draft, Methodology, Investigation. Sofia Tilon: Writing – review & editing, Writing – original draft, Investigation, Conceptualization.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

Acknowledgements

This research has received funding from Government of Spain through RYC2022-038100-I by MCIN/AEI/10.13039/501100011033 and FSE+, PID2021-123475OA-I00 by MCIN/AEI/10.13039/ 501100011033/ FEDER, UE; and Xunta de Galicia - GAIN ED431F 2022/08. This paper was carried out in the frame- work of the SUM4Re project (Creating materials banks from digital urban mining), which has received funding from the Horizon Europe re- search and innovation program under grant agreement no. 101129961. Funded by the European Union. Views and opinions expressed are those of the authors only and do not necessarily reflect those of the EU or HADEA. Neither the EU nor the granting authority can be held responsible for them. Funding for

open access charge: Universidade de Vigo/CISUG.

References

- S. Albers, J. Daft, S. Stabenow, V. Rundshagen, The long-haul low-cost airline business model: A disruptive innovation perspective, J. Air. Transp. Manage 89 (2020) 101878, https://doi.org/10.1016/j.jairtraman.2020.101878.
- [2] J.N. Balls, Low-cost, adaptable solutions sell: Re-thinking off-grid solar diffusion at the bottom of the pyramid in India, Energy Res. Soc. Sci. 70 (2020) 101811, https://doi.org/10.1016/j.erss.2020.101811.
- [3] L.L. Wald, P.C. McDaniel, T. Witzel, J.P. Stockmann, C.Z. Cooley, Low-cost and portable MRI, J. Magnet. Resonance Imag. 52 (2020) 686–696, https://doi.org/ 10.1002/jmri.26942.
- [4] Y.-C. Huang, Low-cost airlines not so low-cost exploring the relationships among travel motivation, service quality and satisfaction: the moderating roles of perceived value, Res. Transport. Business Manage. 49 (2023) 101008, https:// doi.org/10.1016/j.rtbm.2023.101008.
- [5] H. García-Martínez, E. Ávila-Navarro, G. Torregrosa-Penalva, A. Rodríguez-Martínez, C. Blanco-Angulo, M. de la Casa-Lillo, Low-Cost Additive Manufacturing Techniques Applied to the Design of Planar Microwave Circuits by Fused Deposition Modeling, Polymers. (Basel) 12 (2020), https://doi.org/ 10.3390/polym12091946.
- [6] L. Joseph, B.-M. Jun, J.R.V. Flora, C.M. Park, Y. Yoon, Removal of heavy metals from water sources in the developing world using low-cost materials: A review, Chemosphere 229 (2019) 142–159, https://doi.org/10.1016/j. chemosphere.2019.04.198.
- [7] P.B. Aurtherson, B.T. Nalla, K. Srinivasan, K. Mehar, Y. Devarajan, Biofuel production from novel Prunus domestica kernel oil: process optimization technique, BioMass Convers. Biorefin. 13 (2023) 6249–6255, https://doi.org/ 10.1007/s13399-021-01551-5.
- [8] I. Hernández-Neuta, F. Neumann, J. Brightmeyer, T. Ba Tis, N. Madaboosi, Q. Wei, A. Ozcan, M. Nilsson, Smartphone-based clinical diagnostics: towards democratization of evidence-based health care, J. Intern. Med. 285 (2019) 19–39, https://doi.org/10.1111/joim.12820.
- [9] T. Murni, The Effect of Entrepreneurial Orientation to Low Cost Strategy, Differentiation Strategy, Sustainable Innovation and Performance of Small and Medium Enterprises (Studies at Batik Small and Medium Enterprises in East Java Province, Indonesia), European Journal of Business and Management 9 (2017) 137–153.
- [10] B. Sharp, J. Dawes, What is Differentiation and How Does it Work ? Journal of Marketing Management 17 (2001) 739–759, https://doi.org/10.1362/ 026725701323366809.
- [11] K.J. Mason, Marketing low-cost airline services to business travellers, J. Air. Transp. Manage 7 (2001) 103–109, https://doi.org/10.1016/S0969-6997(00) 00036-3.
- [12] S. He, H.-S. Shin, S. Xu, A. Tsourdos, Distributed estimation over a low-cost sensor network: A Review of state-of-the-art, Information Fusion 54 (2020) 21–43, https://doi.org/10.1016/j.inffus.2019.06.026.
- [13] S.I. Granshaw, Photogrammetric terminology: fourth edition, The Photogrammetric Record 35 (2020) 143–288, https://doi.org/10.1111/ phor.12314.
- [14] J. Shan, C.K. Toth, Topographic laser ranging and scanning: principles and processing, CRC press, 2018.
- [15] C.L. Glennie, P.J. Hartzell, Accuracy assessment and calibration of low-cost autonomous LiDAR sensors, Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci. XLIII-B1-2020, 2020, pp. 371–376, https://doi.org/10.5194/isprs-archives-XLIII-B1-2020-371-2020.
- [16] M.S. Mohammed, A.M. Abduljabar, M.M. Faisal, B.M. Mahmmod, S. H. Abdulhussain, W. Khan, P. Liatsis, A. Hussain, Low-cost autonomous car level 2: Design and implementation for conventional vehicles, Results. Eng. 17 (2023) 100969, https://doi.org/10.1016/j.rineng.2023.100969.
- [17] R. Otero, S. Lagüela, I. Garrido, P. Arias, Mobile indoor mapping technologies: A review, Autom. Constr. 120 (2020) 103399, https://doi.org/10.1016/j. autcon.2020.103399.
- [18] M. Elhashash, H. Albanwan, R. Qin, A Review of Mobile Mapping Systems: From Sensors to Applications, Sensors 22 (2022), https://doi.org/10.3390/s22114262.
- [19] S.A. Bello, S. Yu, C. Wang, J.M. Adam, J. Li, Review: Deep Learning on 3D Point Clouds, Remote Sens. (Basel) 12 (2020), https://doi.org/10.3390/rs12111729.
- [20] Y. Guo, H. Wang, Q. Hu, H. Liu, L. Liu, M. Bennamoun, Deep Learning for 3D Point Clouds: A Survey, IEEe Trans. Pattern. Anal. Mach. Intell. 43 (2021) 4338–4364, https://doi.org/10.1109/TPAMI.2020.3005434.
- [21] H. Zhang, C. Wang, S. Tian, B. Lu, L. Zhang, X. Ning, X. Bai, Deep learning-based 3D point cloud classification: A systematic survey and outlook, Displays 79 (2023) 102456, https://doi.org/10.1016/j.displa.2023.102456.
- [22] Y. Xie, J. Tian, X.X. Zhu, Linking Points With Labels in 3D: A Review of Point Cloud Semantic Segmentation, IEEe Geosci. Remote Sens. Mag. 8 (2020) 38–59, https://doi.org/10.1109/MGRS.2019.2937630.
- [23] J. Cheng, L. Zhang, Q. Chen, X. Hu, J. Cai, A review of visual SLAM methods for autonomous driving vehicles, Eng. Appl. Artif. Intell. 114 (2022) 104992, https:// doi.org/10.1016/j.engappai.2022.104992.
- [24] S. Bi, C. Yuan, C. Liu, J. Cheng, W. Wang, Y. Cai, A Survey of Low-Cost 3D Laser Scanning Technology, Appl. Sci. 11 (2021), https://doi.org/10.3390/ app11093938.
- [25] D.V. Nam, K. Gon-Woo, Solid-State LiDAR based-SLAM: A Concise Review and Application, in: 2021 IEEE International Conference on Big Data and Smart

Computing (BigComp), 2021, pp. 302–305, https://doi.org/10.1109/ BigComp51126.2021.00064.

- [26] J.A. Ariza, J.M. Pearce, Low-Cost Assistive Technologies for disabled people using open-source hardware and software: a systematic literature review, IEEe Access. 10 (2022) 124894–124927, https://doi.org/10.1109/ACCESS.2022.3221449.
- [27] C.L. Frush, A new lidar signal processor using digital techniques to provide realtime display, Opt. Quantum. Electron. 7 (1975) 179–185, https://doi.org/ 10.1007/BF00619590.
- [28] N.J. van Eck, L. Waltman, Software survey: VOSviewer, a computer program for bibliometric mapping, Scientometrics. 84 (2010) 523–538, https://doi.org/ 10.1007/s11192-009-0146-3.
- [29] J. Gu, Y. Wang, L. Chen, Z. Zhao, Z. Xuanyuan, K. Huang, A Reliable Road Segmentation and Edge Extraction for Sparse 3D Lidar Data, in: 2018 IEEE Intelligent Vehicles Symposium (IV), IEEE, Changshu, 2018, pp. 1452–1457, https://doi.org/10.1109/IVS.2018.8500486.
- [30] J. Li, B. Yang, Y. Cong, L. Cao, X. Fu, Z. Dong, 3D Forest Mapping Using A Low-Cost UAV Laser Scanning System: Investigation and Comparison, Remote Sens. (Basel) 11 (2019) 717, https://doi.org/10.3390/rs11060717.
- [31] K. Wang, N. Jiasheng, L. Yanqiang, A Robust LiDAR State Estimation and Map Building Approach for Urban Road, in: 2021 IEEE 2nd International Conference on Big Data, Artificial Intelligence and Internet of Things Engineering (ICBAIE), IEEE, Nanchang, China, 2021, pp. 502–506, https://doi.org/10.1109/ ICBAIE52039.2021.9390023.
- [32] M. Soilán, D. González-Aguilera, A. del-Campo-Sánchez, D. Hernández-López, S. Del Pozo, Road marking degradation analysis using 3D point cloud data acquired with a low-cost Mobile Mapping System, Autom. Constr. 141 (2022) 104446, https://doi.org/10.1016/j.autcon.2022.104446.
- [33] A. Ganji, O. Youssefi, J. Xu, K. Mallinen, M. Lloyd, A. Wang, A. Bakhtari, S. Weichenthal, M. Hatzopoulou, Design, calibration, and testing of a mobile sensor system for air pollution and built environment data collection: The urban scanner platform, Environmental Pollution 317 (2023) 120720, https://doi.org/ 10.1016/j.envpol.2022.120720.
- [34] T. Hu, X. Sun, Y. Su, H. Guan, Q. Sun, M. Kelly, Q. Guo, Development and Performance Evaluation of a Very Low-Cost UAV-Lidar System for Forestry Applications, Remote Sens. (Basel) 13 (2021), https://doi.org/10.3390/ rs13010077.
- [35] J. Li, B. Yang, Y. Yang, X. Zhao, Y. Liao, N. Zhu, W. Dai, R. Liu, R. Chen, Z. Dong, Real-time automated forest field inventory using a compact low-cost helmetbased laser scanning system, Int. J. Appl. Earth Observ. Geoinf. 118 (2023) 103299, https://doi.org/10.1016/j.jag.2023.103299.
- [36] A. Rouzbeh Kargar, R.A. MacKenzie, M. Apwong, E. Hughes, J. van Aardt, Stem and root assessment in mangrove forests using a low-cost, rapid-scan terrestrial laser scanner, Wetl. Ecol. Manage 28 (2020) 883–900, https://doi.org/10.1007/ s11273-020-09753-w.
- [37] Z. Wang, X. Zhao, Z. Xu, Offline mapping for autonomous vehicles with low-cost sensors, Computers & Electrical Engineering 82 (2020) 106552, https://doi.org/ 10.1016/j.compeleceng.2020.106552.
- [38] G. Ariante, U. Papa, S. Ponte, G. Del Core, UAS for positioning and field mapping using LIDAR and IMU sensors data: Kalman filtering and integration, in: 2019 IEEE 5th International Workshop on Metrology for AeroSpace (MetroAeroSpace), 2019, pp. 522–527, https://doi.org/10.1109/MetroAeroSpace.2019.8869696.
- [39] J. Li, B. Yang, C. Chen, W. Wu, L. Zhang, Aerial-triangulation aided boresight calibration for a low-cost UAV-lidar system, ISPRS Annals of the Photogrammetry, Remote Sens. Spatial Inf. Sci. V-1–2020 (2020) 245–252, https://doi.org/10.5194/isprs-annals-V-1-2020-245-2020.
- [40] B. Yang, J. Li, A hierarchical approach for refining point cloud quality of a low cost UAV LiDAR system in the urban environment, ISPRS J. Photogr. Remote Sens. 183 (2022) 403–421, https://doi.org/10.1016/j.isprsjprs.2021.11.022.
- [41] Z. Qiu, J. Martínez-Sánchez, P. Arias, M. Datcu, A novel low-cost multi-sensor solution for pavement distress segmentation and characterization at night, Int. J. Appl. Earth Observ. Geoinf. 120 (2023) 103331, https://doi.org/10.1016/j. jag.2023.103331.
- [42] G.Y. Çakir, C.J. Post, E.A. Mikhailova, M.A. Schlautman, 3D LiDAR scanning of urban forest structure using a consumer tablet, Urban Science 5 (2021), https:// doi.org/10.3390/urbansci5040088.
- [43] C. Gollob, T. Ritter, R. Kraßnitzer, A. Tockner, A. Nothdurft, Measurement of forest inventory parameters with Apple iPad Pro and integrated LiDAR technology, Remote Sens. (Basel) 13 (2021), https://doi.org/10.3390/ rs13163129.
- [44] A. Murtiyoso, P. Grussenmeyer, T. Landes, H. Macher, FIRST, Assessments into the use of commercial-grade solid state lidar for low cost heritage documentation, Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci XLIII-B2-2021 (2021) 599–604, https://doi.org/10.5194/isprs-archives-XLIII-B2-2021-599-2021.
- [45] M. Brach, W. Tracz, G. Krok, J. Gasior, Feasibility of Low-Cost LiDAR scanner implementation in forest sampling techniques, Forests. 14 (2023), https://doi. org/10.3390/f14040706.
- [46] Q. Gautier, T. Garrison, F. Rushton, N. Bouck, E. Lo, P. Tueller, C. Schurgers, R. Kastner, Low-cost 3D scanning systems for cultural heritage documentation, J. Cultural Heritage Manage. Sustain. Dev. Ahead-of-print (2020), https://doi. org/10.1108/JCHMSD-03-2020-0032.
- [47] S. Timilsina, J. Aryal, J.B. Kirkpatrick, Mapping urban tree cover changes using object-based convolution neural network (OB-CNN), Remote Sens. (Basel) 12 (2020), https://doi.org/10.3390/rs12183017.
- [48] S. Dayal, S. Goel, B. Lohani, N. Mittal, R. Mishra, Comprehensive airborne laser scanning (ALS) simulation, J. Ind. Soc. Remote Sens. 49 (2021), https://doi.org/ 10.1007/s12524-021-01334-5.

- [49] A. Gressin, J. Vallet, M. Bron, About photogrammetric UAV-mapping: which accuracy for which application?, in: Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci XLIII-B2-2020, 2020, pp. 1081–1089, https://doi.org/10.5194/isprs-archives-XLIII-B2-2020-1081-2020.
- [50] J. Li, B. Yang, Y. Chen, W. Wu, Y. Yang, X. Zhao, R. Chen, Evaluation of a compact helmet-based laser scanning system for aboveground and underground 3D mapping, in: Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci XLIII-B2-2022, 2022, pp. 215–220, https://doi.org/10.5194/isprs-archives-XLIII-B2-2022-215-2022.
- [51] G. Ariante, U. Papa, S. Ponte, G. Del Core, Real-time obstacle detection and field mapping system using LIDAR-ToF sensors for small UAS, in: G. Di Francia, C. Di Natale (Eds.), Sensors and Microsystems, Springer International Publishing, Cham. 2021. pp. 9–16. https://doi.org/10.1007/978-3-030-69551-4 2.
- Cham, 2021, pp. 9–16, https://doi.org/10.1007/978-3-030-69551-4_2.
 [52] S. Li, X. Li, H. Wang, Y. Zhou, Z. Shen, Multi-GNSS PPP/INS/Vision/LiDAR tightly integrated system for precise navigation in urban environments, Information Fusion 90 (2023) 218–232, https://doi.org/10.1016/j.inffus.2022.09.018.
- [53] S. Li, S. Wang, Y. Zhou, Z. Shen, X. Li, Tightly Coupled Integration of GNSS, INS, and LiDAR for Vehicle Navigation in Urban Environments, IEEe Internet. Things. J. 9 (2022) 24721–24735, https://doi.org/10.1109/JIOT.2022.3194544.
- [54] W. Tan, D. Zhang, L. Ma, L. Wang, N. Qin, Y. Chen, J. Li, Semantic Segmentation of UAV Lidar Point Clouds of a Stack Interchange with Deep Neural Networks, in: 2021 IEEE International Geoscience and Remote Sensing Symposium IGARSS, IEEE, Brussels, Belgium, 2021, pp. 582–585, https://doi.org/10.1109/ IGARSS47720.2021.9554610.
- [55] F. Vultaggio, F. d'Apolito, C. Sulzbachner, P. Fanta-Jende, Simulation of low-cost mems-lidar and analysis of its effect on the performances of state-of-the-art slams, Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci XLVIII-1/W1-2023 (2023) 539–545, https://doi.org/10.5194/isprs-archives-XLVIII-1-W1-2023-539-2023.
- [56] A. Elamin, A. El-Rabbany, UAV-Based Multi-Sensor Data fusion for urban land cover mapping using a deep convolutional neural network, Remote Sens. (Basel) 14 (2022) 4298, https://doi.org/10.3390/rs14174298.
- [57] N. Jeong, H. Hwang, E.T. Matson, Evaluation of low-cost LiDAR sensor for application in indoor UAV navigation, in: 2018 IEEE Sensors Applications Symposium (SAS), IEEE, Seoul, Korea (South), 2018, pp. 1–5, https://doi.org/ 10.1109/SAS.2018.8336719.
- [58] S. Parve, O. Gaikwad, P. Sidhapura, S. Chaudhary, S. Nakhye, Development of low-cost LiDAR scanner for indoor mapping, (2021).
- [59] J. Kang, D. Kim, C. Lee, J. Kang, D. Kim, Efficiency Study of Combined UAS Photogrammetry and Terrestrial LiDAR in 3D Modeling for Maintenance and Management of Fill Dams, Remote Sens. (Basel) 15 (2023), https://doi.org/ 10.3390/rs15082026.
- [60] S. Benchekroun, I.I.T. Ullah, Preserving the Past for an Uncertain Future: Accessible, Low-Cost Methods for 3-D Data Creation, Processing, and Dissemination in Digital Cultural Heritage Preservation, in: The 26th International Conference on 3D Web Technology, ACM, Pisa Italy, 2021, pp. 1–9, https://doi.org/10.1145/3485444.3507684.
- [61] L. Chen, X. He, X. Zhao, H. Li, Y. Huang, B. Zhou, W. Chen, Y. Li, C. Wen, C. Wang, GoComfort: Comfortable Navigation for Autonomous Vehicles Leveraging High-Precision Road Damage Crowdsensing, IEEE Trans. on Mobile Comput. (2022) 1–18, https://doi.org/10.1109/TMC.2022.3198089.
- [62] B. Bessell, A. Singh, A low-cost open-source handheld LiDAR-based automated understory timber stand surveying device, HardwareX. 12 (2022) e00339, https://doi.org/10.1016/j.ohx.2022.e00339.
- [63] S. Zhang, F. Han, S.M. Bogus, Building Footprint and Height Information Extraction from Airborne LiDAR and Aerial Imagery, in: Construction Research Congress 2020, American Society of Civil Engineers, Tempe, Arizona, 2020, pp. 326–335, https://doi.org/10.1061/9780784482865.035.
- [64] Q. Zhao, L. Zhou, G. Lv, A 3D modeling method for buildings based on LiDAR point cloud and DLG, Computers, Environment and Urban Systems 102 (2023) 101974, https://doi.org/10.1016/j.compenvurbsys.2023.101974.
- [65] E. Aldao, H. González-Jorge, L.M. González-deSantos, G. Fontenla-Carrera, J. Martínez-Sánchez, Validation of Solid-State LiDAR Measurement System for Ballast Geometry Monitoring in Rail Tracks, Infrastructures. (Basel) 8 (2023) 63, https://doi.org/10.3390/infrastructures8040063.
- [66] Y. Ren, C. Ai, P. Lu, Z. Dai, H. Wang, An Automated Rail Extraction Framework for Low-Density LiDAR Data Without Sensor Configuration Information, IEEE Sensors J 22 (2022) 13234–13243, https://doi.org/10.1109/ JSEN.2022.3177698.
- [67] S. Mehtab, W.Q. Yan, A. Narayanan, 3D Vehicle Detection Using Cheap LiDAR and Camera Sensors, in: 2021 36th International Conference on Image and Vision Computing New Zealand (IVCNZ), IEEE, Tauranga, New Zealand, 2021, pp. 1–6, https://doi.org/10.1109/IVCNZ54163.2021.9653358.
- [68] H. Domínguez, A. Morcillo, M. Soilán, D. González-Aguilera, Automatic recognition and geolocation of vertical traffic signs based on artificial intelligence using a low-cost mapping mobile system, Infrastructures. (Basel) 7 (2022) 133, https://doi.org/10.3390/infrastructures7100133.
- [69] M.L.R. Lagahit, M. Matsuoka, Focal combo loss for improved road marking extraction of sparse mobile lidar scanning point cloud-derived images using convolutional neural networks, Remote Sens. (Basel) 15 (2023) 597, https://doi. org/10.3390/rs15030597.
- [70] H. Tardy, M. Soilán, J.A. Martín-Jiménez, D. González-Aguilera, Automatic road inventory using a low-cost mobile mapping system and based on a semantic segmentation deep learning model, Remote Sens. (Basel) 15 (2023) 1351, https:// doi.org/10.3390/rs15051351.

- [71] R. Yang, Q. Li, J. Tan, S. Li, X. Chen, Accurate Road Marking Detection from Noisy Point Clouds Acquired by Low-Cost Mobile LiDAR Systems, IJGI 9 (2020) 608, https://doi.org/10.3390/ijgi9100608.
- [72] A. Bhatt, S. Bharadwaj, V.B. Sharma, R. Dubey, S. Biswas, An overview of road health monitoring system for rigid pavement by terrestrial laser scanner, Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci XLIII-B1-2022 (2022) 173–180, https://doi.org/10.5194/isprs-archives-XLIII-B1-2022-173-2022.
- [73] P. Cao, Y. Wang, X. Liu, Trajectory-Based Method for Dividing Lanes of Vehicle Trajectories Collected by Roadside LiDAR, Transportation Research Record: Journal of the Transportation Research Board (2023) 03611981231182987, https://doi.org/10.1177/03611981231182987.
- [74] V. Kumar, S.C. Subramanian, R. Rajamani, A novel algorithm to track closely spaced road vehicles using a low density flash lidar, Signal. Processing. 191 (2022) 108360, https://doi.org/10.1016/j.sigpro.2021.108360.
- [75] A.E.L. Stovall, J.W. Atkins, Assessing Low-Cost Terrestrial Laser Scanners for Deriving Forest Structure Parameters, EARTH SCIENCES (2021), https://doi.org/ 10.20944/preprints202107.0690.v1.
- [76] S. Tilon, F. Nex, G. Vosselman, I. Sevilla de la Llave, N. Kerle, Towards Improved Unmanned Aerial Vehicle Edge Intelligence: A Road Infrastructure Monitoring Case Study, Remote Sens. (Basel) 14 (2022), https://doi.org/10.3390/ rs14164008.
- [77] B. Essel, M. Bolger, J. McDonald, C. Cahalane, ADGEO: A new shore-based approach to improving spatial accuracy when mapping water bodies using lowcost drones, The Photogrammetric Record (2024), https://doi.org/10.1111/ phor.12512 n/a.
- [78] F. Gaspari, F. Ioli, F. Barbieri, E. Belcore, L. Pinto, Integration of UAV-lidar and UAV-photogrammetry for infrastructure monitoring and bridge assessment, Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci. XLIII-B2-2022 (2022) 995–1002, https://doi.org/10.5194/isprs-archives-XLIII-B2-2022-995-2022.
- [79] T. Bakirman, B. Bayram, B. Akpinar, M.F. Karabulut, O.C. Bayrak, A. Yigitoglu, D. Z. Seker, Implementation of ultra-light UAV systems for cultural heritage documentation, J. Cult. Herit. 44 (2020) 174–184, https://doi.org/10.1016/j. culher.2020.01.006.
- [80] R. Garozzo, Survey, semantics, and H-BrIM modeling for health assessment of masonry railroad bridges, (2023).
- [81] Kyriacos Themistocleous, C. Mettas, E. Evagorou, D. Hadjimitsis, The use of UAVs and photogrammetry for the documentation of cultural heritage monuments: the case study of the churches in Cyprus, in: 2019: p. 111560I. https://doi.org /10.1117/12.2533056.
- [82] S. Manajitprasert, N.K. Tripathi, S. Arunplod, Three-dimensional (3D) modeling of cultural heritage site using uav imagery: a case study of the pagodas in Wat Maha That, Thailand, Applied Sciences 9 (2019), https://doi.org/10.3390/ app9183640.
- [83] A. Ulvi, Documentation, three-dimensional (3D) modelling and visualization of cultural heritage by using unmanned aerial vehicle (UAV) photogrammetry and terrestrial laser scanners, Int. J. Remote Sens. 42 (2021) 1994–2021, https://doi. org/10.1080/01431161.2020.1834164.
- [84] S. Karam, G. Vosselman, M. Peter, S. Hosseinyalamdary, V. Lehtola, Design, calibration, and evaluation of a backpack indoor mobile mapping system, Remote Sens. (Basel) 11 (2019), https://doi.org/10.3390/rs11080905.
- [85] Y. Su, H. Guan, T. Hu, Q. Guo, The integration of Uavand backpack lidar systems for forest inventory, in: IGARSS 2018 - 2018 IEEE International Geoscience and Remote Sensing Symposium, IEEE, Valencia, 2018, pp. 8757–8760, https://doi. org/10.1109/IGARSS.2018.8517639.
- [86] J. Balado, E. Frías, S.M. González-Collazo, L. Díaz-Vilariño, New Trends in Laser Scanning for Cultural Heritage, in: D. Bienvenido-Huertas, J. Moyano-Campos (Eds.), New technologies in building and construction: towards sustainable development, Springer Nature Singapore, Singapore, 2022: pp. 167–186. https ://doi.org/10.1007/978-981-19-1894-0_10.
- [87] J. McGlade, L. Wallace, B. Hally, K. Reinke, S. Jones, The effect of surrounding vegetation on basal stem measurements acquired using low-cost depth sensors in urban and native forest environments, Sensors 23 (2023) 3933, https://doi.org/ 10.3390/s23083933.
- [88] S. Tatsumi, K. Yamaguchi, N. Furuya, ForestScanner: A mobile application for measuring and mapping trees with LiDAR -equipped iPhone and iPad, Methods Ecol. Evol. 14 (2023) 1603–1609, https://doi.org/10.1111/2041-210X.13900.
- [89] S. Hallegatte, J. Rentschler, J. Rozenberg, Lifelines: The resilient infrastructure opportunity, World Bank Publications, 2019.
- [90] R.S. Panah, M. Kioumarsi, Application of building information modelling (BIM) in the health monitoring and maintenance process: a systematic review, Sensors 21 (2021), https://doi.org/10.3390/s21030837.
- [91] A. Elamin, A. El-Rabbany, UAV-Based Image and LiDAR Fusion for Pavement Crack Segmentation, Sensors 23 (2023) 9315, https://doi.org/10.3390/ s23239315.
- [92] W. Tan, D. Zhang, L. Ma, L. Wang, N. Qin, Y. Chen, J. Li, Semantic segmentation of UAV lidar point clouds of a stack interchange with deep neural networks, in: 2021 IEEE International Geoscience and Remote Sensing Symposium IGARSS, IEEE, Brussels, Belgium, 2021, pp. 582–585, https://doi.org/10.1109/ IGARS547720.2021.9554610.
- [93] C. Torresan, A. Berton, F. Carotenuto, U. Chiavetta, F. Miglietta, A. Zaldei, B. Gioli, Development and performance assessment of a low-cost UAV laser scanner system (LasUAV), Remote Sens. (Basel) 10 (2018) 1094, https://doi.org/ 10.3390/rs10071094.
- [94] L. Mot, S. Hong, K. Charoenjit, H. Zhang, Tree height estimation using field measurement and low-cost unmanned aerial vehicle (UAV) at Phnom Kulen National Park of Cambodia, in: 2021 9th International Conference on Agro-

Geoinformatics (Agro-Geoinformatics), IEEE, Shenzhen, China, 2021, pp. 1–4, https://doi.org/10.1109/Agro-Geoinformatics50104.2021.9530357.

- [95] X. Li, S. Wang, S. Li, Y. Zhou, C. Xia, Z. Shen, Enhancing RTK performance in urban environments by tightly integrating INS and LiDAR, IEEE Trans. Veh. Technol. 72 (2023) 9845–9856, https://doi.org/10.1109/TVT.2023.3257874.
- [96] M. Rahman, H. Liu, M. Masri, I. Durazo-Cardenas, A. Starr, A railway track reconstruction method using robotic vision on a mobile manipulator: A proposed strategy, Comput. Ind. 148 (2023) 103900, https://doi.org/10.1016/j. compind.2023.103900.
- [97] Y. Cong, C. Chen, B. Yang, J. Li, W. Wu, Y. Li, Y. Yang, 3D-CSTM: A 3D continuous spatio-temporal mapping method, ISPRS Journal of Photogrammetry and Remote Sensing 186 (2022) 232–245, https://doi.org/10.1016/j.isprsjprs.2022.02.005.
- [98] D. Tu, H. Cui, S. Shen, PanoVLM: Low-Cost and accurate panoramic vision and LiDAR fused mapping, ISPRS Journal of Photogrammetry and Remote Sensing 206 (2023) 149–167, https://doi.org/10.1016/j.isprsjprs.2023.11.012.
- [99] L. Jurjević, X. Liang, M. Gašparović, I. Balenović, Is field-measured tree height as reliable as believed – Part II, A comparison study of tree height estimates from conventional field measurement and low-cost close-range remote sensing in a deciduous forest, ISPRS Journal of Photogrammetry and Remote Sensing 169 (2020) 227–241, https://doi.org/10.1016/j.isprsjprs.2020.09.014.
- [100] L. Teppati Losè, A. Spreafico, F. Chiabrando, F. Giulio Tonolo, Apple LiDAR Sensor for 3D Surveying: Tests and Results in the Cultural Heritage Domain, Remote Sens. (Basel) 14 (2022), https://doi.org/10.3390/rs14174157.
- [101] A. Masiero, G. Tucci, A. Vettore, Tree Detection with a Mobile Laser Scanner, IOP Conf. Ser.: Earth Environ. Sci. 767 (2021) 012034, https://doi.org/10.1088/ 1755-1315/767/1/012034.
- [102] B.-K.D. Park, J.R. Sayer, A.D. Clover, M.P. Reed, Longitudinal degradation of pavement marking detectability for mobile lidar sensing technology in real-world use, Sensors 23 (2023) 5815, https://doi.org/10.3390/s23135815.
- [103] Z. Qiu, J. Martínez-Sánchez, V.M. Brea, P. López, P. Arias, Low-cost mobile mapping system solution for traffic sign segmentation using Azure Kinect, Int. J. Appl. Earth Observ. Geoinf. 112 (2022) 102895, https://doi.org/10.1016/j. jag.2022.102895.
- [104] G. Yang, Y. Zhao, B. Li, Y. Ma, R. Li, J. Jing, Y. Dian, Tree species classification by employing multiple features acquired from integrated sensors, J. Sens. 2019 (2019) 1–12, https://doi.org/10.1155/2019/3247946.
- [105] R.Q. Charles, H. Su, M. Kaichun, L.J. Guibas, PointNet: Deep Learning on Point Sets for 3D Classification and Segmentation, in: 2017 IEEE Conference on Computer Vision and Pattern Recognition (CVPR, 2017, pp. 77–85, https://doi. org/10.1109/CVPR.2017.16.
- [106] A. Singh, K. Raj, A.M. Roy, Efficient deep learning-based semantic mapping approach using monocular vision for resource-limited mobile robots, J. Intell. Robot. Syst. 109 (2023) 69, https://doi.org/10.1007/s10846-023-01988-y.
- [107] M.L.R. Lagahit, Z. Li, K. Sakaguchi, M. Matsuoka, EXPLORING GROUND SEGMENTATION FROM LIDAR SCANNING-DERIVED IMAGES USING CONVOLUTIONAL NEURAL NETWORKS, Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci XLVIII-1/W1-2023 (2023) 221–226, https://doi.org/10.5194/ isprs-archives-XLVIII-1-W1-2023-221-2023.
- [108] A.-K. Vo, T. Huynh-The, LiPoSeg: A Lightweight Encoder-Decoder Network for LiDAR-based Road-Object Semantic Segmentation, in: Proceedings of the 12th International Symposium on Information and Communication Technology, ACM, Ho Chi Minh Vietnam, 2023, pp. 422–428, https://doi.org/10.1145/ 3628797.3628900.
- [109] Li Fei-Fei, R. Fergus, P. Perona, One-shot learning of object categories, IEEe Trans. Pattern. Anal. Mach. Intell. 28 (2006) 594–611, https://doi.org/10.1109/ TPAMI.2006.79.
- [110] A. Kirillov, E. Mintun, N. Ravi, H. Mao, C. Rolland, L. Gustafson, T. Xiao, S. Whitehead, A.C. Berg, W.-Y. Lo, P. Dollar, R. Girshick, Segment Anything, in: Proceedings of the IEEE/CVF International Conference on Computer Vision (ICCV), 2023, pp. 4015–4026.
- [111] M. Sewtz, Y. Fanger, X. Luo, T. Bodenmüller, R. Triebel, IndoorMCD: A Benchmark for Low-Cost Multi-Camera SLAM in Indoor Environments, IEEe Robot. Autom. Lett. 8 (2023) 1707–1714, https://doi.org/10.1109/ LRA.2023.3236840.
- [112] J. Li, S. Yuan, M. Cao, T.-M. Nguyen, K. Cao, L. Xie, HCTO: Optimality-aware LiDAR inertial odometry with hybrid continuous time optimization for compact wearable mapping system, ISPRS Journal of Photogrammetry and Remote Sensing 211 (2024) 228–243, https://doi.org/10.1016/j.isprsjprs.2024.04.004.
- [113] A. Tobergte, M. Pomarlan, G. Passig, G. Hirzinger, An approach to ulta-tightly coupled data fusion for handheld input devices in robotic surgery, in: 2011 IEEE International Conference on Robotics and Automation, 2011, pp. 2424–2430, https://doi.org/10.1109/ICRA.2011.5980120.
- [114] J. Li, W. Wu, B. Yang, X. Zou, Y. Yang, X. Zhao, Z. Dong, WHU-Helmet: A Helmet-Based Multisensor SLAM Dataset for the Evaluation of Real-Time 3-D Mapping in Large-Scale GNSS-Denied Environments, IEEE Transactions on Geoscience and Remote Sensing 61 (2023) 1–16, https://doi.org/10.1109/TGRS.2023.3275307.
- [115] J. Vynikal, D. Zahradník, Floor plan creation using a low-cost 360° camera, The Photogrammetric Record 38 (2023) 520–536, https://doi.org/10.1111/ phor.12463.
- [116] J. Li, B. Yang, C. Chen, A. Habib, NRLI-UAV: Non-rigid registration of sequential raw laser scans and images for low-cost UAV LiDAR point cloud quality improvement, ISPRS Journal of Photogrammetry and Remote Sensing 158 (2019) 123–145, https://doi.org/10.1016/j.isprsjprs.2019.10.009.
- [117] O. Abdi, J. Uusitalo, J. Pietarinen, A. Lajunen, Evaluation of forest features determining GNSS positioning accuracy of a novel low-cost, mobile RTK system

J. Balado et al.

using LiDAR and TreeNet, Remote Sens. (Basel) 14 (2022) 2856, https://doi.org/10.3390/rs14122856.

- [118] H.J. Persson, G. Ståhl, Characterizing uncertainty in forest remote sensing studies, Remote Sens. (Basel) 12 (2020), https://doi.org/10.3390/rs12030505.
- [119] A. Murtiyoso, P. Grussenmeyer, D. Suwardhi, Technical considerations in lowcost heritage documentation, Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci XLII-2/W17 (2019) 225–232, https://doi.org/10.5194/isprs-archives-XLII-2-W17-225-2019.
- [120] G. Barone, R. Garozzo, G. Russo, C. Santagati, D. Sinitò, M.Y. Spironello, F. Stanco, 3D modeling and augmented reality in education: an effective application for the Museo dei Saperi e delle Mirabilia of the University of Catania, in: G.L. Foresti, A. Fusiello, E. Hancock (Eds.), Image Analysis and Processing -ICIAP 2023 Workshops, Springer Nature Switzerland, Cham, 2024, pp. 449–461.
- [121] S. Nebel, M. Beege, S. Schneider, G.D. Rey, A review of photogrammetry and photorealistic 3D models in education from a psychological perspective, Front. Educ. (Lausanne) 5 (2020). https://www.frontiersin.org/articles/10.33 89/feduc.2020.00144.
- [122] T.R.H. Goodbody, N.C. Coops, J.C. White, Digital Aerial Photogrammetry for Updating Area-Based Forest Inventories: A Review of Opportunities, Challenges, and Future Directions, Curr. For. Rep. 5 (2019) 55–75, https://doi.org/10.1007/ s40725-019-00087-2.
- [123] G. D'Agostino, M. Figuera, G. Russo, M. Galizia, P.M. Militello, Integrated 3D survey for the documentation and visualization of a rock-cut underground built heritage, Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci XLVI-2/WI-2022 (2022) 167–174, https://doi.org/10.5194/isprs-archives-XLVI-2-WI-2022-167-2022.
- [124] N. Bruno, L. Perfetti, F. Fassi, R. Roncella, Photogrammetric survey of narrow spaces in cultural heritage: comparison of two multi-camera approaches, Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci XLVIII-2/W4-2024 (2024) 87–94, https://doi.org/10.5194/isprs-archives-XLVIII-2-W4-2024-87-2024.
- [125] C. Faraci, R.E. Musumeci, M. Marino, A. Ruggeri, L. Carlo, B. Jensen, E. Foti, G. Barbaro, B. Elsaßer, Wave- and current-dominated combined orthogonal flows over fixed rough beds, Cont. Shelf. Res. 220 (2021) 104403, https://doi.org/ 10.1016/j.csr.2021.104403.
- [126] L.A. Eberhard, P. Sirguey, A. Miller, M. Marty, K. Schindler, A. Stoffel, Y. Bühler, Intercomparison of photogrammetric platforms for spatially continuous snow depth mapping, Cryosphere 15 (2021) 69–94, https://doi.org/10.5194/tc-15-69-2021.
- [127] S.R. Rogers, I. Manning, W. Livingstone, Comparing the Spatial Accuracy of Digital Surface Models from Four Unoccupied Aerial Systems: Photogrammetry Versus LiDAR, Remote Sens. (Basel) 12 (2020), https://doi.org/10.3390/ rs12172806.
- [128] A. Murtiyoso, P. Grussenmeyer, Experiments using smartphone-based videogrammetry for low-cost cultural heritage documentation, Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci XLVI-M-1-2021 (2021) 487–491, https://doi.org/ 10.5194/isprs-archives-XLVI-M-1-2021-487-2021.
- [129] A. Torresani, F. Remondino, Videogrammetry vs photogrammetry for heritage 3D reconstruction, Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci. XLII-2/W15 (2019) 1157–1162, https://doi.org/10.5194/isprs-archives-XLII-2-W15-1157-2019.

- [130] A.T. Hassan, D. Fritsch, Integration of laser scanning and photogrammetry in 3D/ 4D cultural heritage preservation–a review, Int. J. Appl. 9 (2019) 16.
- [131] E. Maset, R. Valente, M. Iamoni, M. Haider, A. Fusiello, Integration of photogrammetry and portable mobile mapping technology for 3d modeling of cultural heritage sites: the case study of the Bziza temple, Int. Arch. Photogr. Remote Sens. Spatial Inf. Sci XLIII-B2-2022 (2022) 831–837, https://doi.org/ 10.5194/isprs-archives-XLIII-B2-2022-831-2022.
- [132] M. Yusuf, A. Zaidi, A. Haleem, S. Bahl, M. Javaid, S. Bala Garg, J. Garg, IoT-based low-cost 3D mapping using 2D Lidar for different materials, Materials Today: Proceedings 57 (2022) 942–947, https://doi.org/10.1016/j.matpr.2022.03.161.
- [133] X. Qian, C. Liu, X. Qi, S.-C. Tan, E. Lam, N. Wong, Context-aware transformer for 3D point cloud automatic annotation, in: Proceedings of the Thirty-Seventh AAAI Conference on Artificial Intelligence and Thirty-Fifth Conference on Innovative Applications of Artificial Intelligence and Thirteenth Symposium on Educational Advances in Artificial Intelligence, AAAI Press, 2023, https://doi.org/10.1609/ aaai.v37i2.25301.
- [134] K.-C. Hsu, H.-W. Tseng, Accelerating applications using edge tensor processing units, in: Proceedings of the International Conference for High Performance Computing, Networking, Storage and Analysis, Association for Computing Machinery, New York, NY, USA, 2021, https://doi.org/10.1145/ 3458817.3476177.
- [135] M. Eggers, T. Khuon, Neural network data fusion concepts and application, in: 1990 IJCNN International Joint Conference on Neural Networks 2, 1990, pp. 7–16, https://doi.org/10.1109/IJCNN.1990.137687, vol.
- [136] D. Stathas, J.P. Wang, H.I. Ling, Model geogrids and 3D printing, Geotextiles Geomembr. 45 (2017) 688–696, https://doi.org/10.1016/j. geotexmem.2017.07.006.
- [137] A. Kendall, H. Martirosyan, S. Dasgupta, P. Henry, R. Kennedy, A. Bachrach, A. Bry, End-To-End Learning of Geometry and Context for Deep Stereo Regression, in: Proceedings of the IEEE International Conference on Computer Vision (ICCV, 2017.
- [138] Y. Yao, Z. Luo, S. Li, T. Fang, L. Quan, MVSNet: Depth Inference for Unstructured Multi-view Stereo, in: Proceedings of the European Conference on Computer Vision (ECCV), 2018.
- [139] J. Liu, J. Gao, S. Ji, C. Zeng, S. Zhang, J. Gong, Deep learning based multi-view stereo matching and 3D scene reconstruction from oblique aerial images, ISPRS J. Photogr. Remote Sens. 204 (2023) 42–60, https://doi.org/10.1016/j. isprsjprs.2023.08.015.
- [140] V. Croce, D. Billi, G. Caroti, A. Piemonte, L. De Luca, P. Véron, Comparative assessment of neural radiance fields and photogrammetry in digital heritage: impact of varying image conditions on 3D reconstruction, Remote Sens. (Basel) 16 (2024), https://doi.org/10.3390/rs16020301.
- [141] J. Zbontar, Y. LeCun, Computing the stereo matching cost with a convolutional neural network, in: Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition (CVPR), 2015.
- [142] B. Mildenhall, P.P. Srinivasan, M. Tancik, J.T. Barron, R. Ramamoorthi, R. Ng, NeRF: representing scenes as neural radiance fields for view synthesis, Commun. ACM 65 (2021) 99–106, https://doi.org/10.1145/3503250.
- [143] B. Kerbl, G. Kopanas, T. Leimkuehler, G. Drettakis, 3D gaussian splatting for realtime radiance field rendering, ACM Trans. Graph 42 (2023), https://doi.org/ 10.1145/3592433.